Author’s response to reviews

Title: Glomangiomyoma of the neck in a child in Nepal : a rare case report and literature review

Authors:

Bishow Tulachan (tulachanbishow@hotmail.com)
Buddha Borgohain (drborgohain@yahoo.com)
Ravi Swar (ravi.drsvar@gmail.com)
Roshan Acharya (iroshanacharya@gmail.com)
Abishesh Shakya (abishesh.shakya1234511@gmail.com)

Version: 2 Date: 18 Jan 2017

Author’s response to reviews:

As per the editor's suggestion, the identity of the patient in radiological pictures have been concealed.